

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hakenes, Hendrik; Schnabel, Isabel

## Working Paper Credit risk transfer and bank competition

Preprints of the Max Planck Institute for Research on Collective Goods, No. 2009,33

**Provided in Cooperation with:** Max Planck Institute for Research on Collective Goods

*Suggested Citation:* Hakenes, Hendrik; Schnabel, Isabel (2009) : Credit risk transfer and bank competition, Preprints of the Max Planck Institute for Research on Collective Goods, No. 2009,33, Max Planck Institute for Research on Collective Goods, Bonn

This Version is available at: https://hdl.handle.net/10419/32255

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU

Preprints of the Max Planck Institute for Research on Collective Goods Bonn 2009/33



# Credit Risk Transfer and Bank Competition

Hendrik Hakenes Isabel Schnabel



MAX PLANCK SOCIETY



# **Credit Risk Transfer and Bank Competition**

Hendrik Hakenes / Isabel Schnabel

October 2009

Max Planck Institute for Research on Collective Goods, Kurt-Schumacher-Str. 10, D-53113 Bonn http://www.coll.mpg.de

# Credit Risk Transfer and Bank Competition<sup> $\dagger$ </sup>

Hendrik Hakenes<sup>\*</sup> University of Hannover and MPI Bonn

Isabel Schnabel<sup>\*\*</sup> University of Mainz, CEPR, and MPI Bonn

September 28, 2009

Abstract: We present a banking model with imperfect competition in which borrowers' access to credit is improved when banks are able to transfer credit risks. However, the market for credit risk transfer (CRT) works smoothly only if the quality of loans is public information. If the quality of loans is private information, banks have an incentive to grant unprofitable loans in order to transfer them to other parties, leading to an increase in aggregate risk. Nevertheless, the introduction of CRT generally increases welfare in our setup. However, under private information, higher competition induces an expansion of loans to unprofitable firms, which in the limit offsets the welfare gains from CRT completely.

**Keywords:** Credit risk transfer, credit derivatives, public and private information, access to credit, bank competition.

#### JEL Classification: G21, L11, G13.

<sup>†</sup>An earlier version of this paper was circulated under the title "Credit Risk Transfer in Banking Markets with Hard and Soft Information." We would like to thank John Boyd, Florian Heider, Martin Hellwig, Christian Laux, Gyöngyi Lóránth, Alexander Morell, Javier Suarez, Jean Tirole, and participants at the workshop on "Risk transfer mechanisms and financial stability" in Basel, at the ESSFM in Gerzensee, at the ESEM in Milan, the VfS in Graz, the DGF in Münster, the SFBI in Karlsruhe, the Bundesbank-CEPR-CFS Conference on "Risk Transfer: Challenges for Financial Institutions and Markets", the FIRS Conference in Prague, and seminar participants in Bonn, Frankfurt, Vallendar, and Hannover for helpful suggestions.

\*Address: Institute of Financial Economics, Leibniz University of Hannover, Königsworther Platz 1, 30167 Hannover, Germany, +49-511-762-5225, hakenes@fmt.uni-hannover.de.

\*\*Address: Department of Law and Economics, Johannes Gutenberg University Mainz, 55099 Mainz, Germany, +49-6131-39-24191, isabel.schnabel@uni-mainz.de.

# 1 Introduction

In the years before the subprime crisis, many countries have seen an explosion in the use of instruments for credit risk transfer (CRT) by financial institutions. At that time, this development was welcomed by many observers. Most prominently, it was argued that CRT leads to a desirable redistribution and better diversification of credit risks (see, e. g., Allen and Gale, 2005). Another advantage is the potential of CRT to improve the access to credit for firms and households (or, put differently, the ability of banks to free up capital; see, e. g., Chiesa, 2008).<sup>1</sup> However, the advent of the subprime crisis has raised doubts about the overall benefits of credit risk transfer. The recent experience suggests that CRT may also lead to a deterioration of loan quality, with detrimental consequences for financial stability.

From a theoretical perspective, this decline in loan quality did not come unexpectedly. The early literature on credit risk transfer emphasized the reduced monitoring incentives of banks, once a loan has been transferred to a third party (see, e. g., Pennacchi, 1988; Gorton and Pennacchi, 1995).<sup>2</sup> However, recent empirical findings also suggest that there has been an expansion of low quality loans.<sup>3</sup> Many of the loans granted during the credit boom preceding the subprime crisis were of such a bad quality that banks must have been aware of the poor loan quality when the loan was granted (an extreme example are the notorious "ninja" loans). It seems that banks granted low quality loans and transferred them to other parties afterwards.<sup>4</sup>

In addition, the decrease in lending standards on the eve of the subprime crisis has been shown to be related to the market structure in the banking sector. Dell'Ariccia,

<sup>&</sup>lt;sup>1</sup>For an excellent survey on credit risk transfer, see Duffie (2007).

<sup>&</sup>lt;sup>2</sup>Other papers dealing with the effects of CRT on monitoring incentives include Morrison (2005), Chiesa (2008), Parlour and Plantin (2008), and Cerasi and Rochet (2008). See Ashcraft and Santos (2008) for empirical evidence.

<sup>&</sup>lt;sup>3</sup>Dell'Ariccia, Igan, and Laeven (2008) document a decline in loan denial rates, which they interpret as a decrease in lending standards.

<sup>&</sup>lt;sup>4</sup>This view is supported by the finding of Dell'Ariccia, Igan, and Laeven (2008) that the decline in loan denial rates was more pronounced in regions with higher securitization rates. Moreover, Keys, Mukherjee, Seru, and Vig (2009a) show that loans eligible for securitization on average defaulted much more frequently than loans with similar observable risk characteristics that were not eligible for CRT. They interpret their finding as evidence for laxer screening of loans that were to be securitized.

Igan, and Laeven (2008) show that loan denial rates in the subprime segment decreased more in areas with highly competitive banking markets and that the market entry of new financial institutions induced a further decrease in lending standards. The role of banking competition in the presence of credit risk transfer has to our knowledge not yet been dealt with in the theoretical literature.

Our paper models banks' moral hazard problem in the *origination* of loans and shows how it is affected by the degree of competition in the banking sector. We start from a banking model with imperfect competition, in which the access of risky, but profitable borrowers to bank credit is constrained due to banks' limited risk-bearing capacities. Such constraints may arise from regulatory constraints, bankruptcy costs, or bankers' risk aversion. We show that the credit constraints are especially tight if banking markets are highly competitive. The reason is that the rents from relatively safe loans, which can serve as a buffer for riskier activities, will be small in the presence of fierce competition.

We then show that such credit constraints may be relaxed by allowing banks to transfer risks to outside investors. However, the functioning of CRT markets depends crucially on the type of information on which bank loans are based. If loans are granted on the basis of publicly observable information, a transfer of credit risk works smoothly and the access to credit for risky, but profitable borrowers is improved. Since the information is public, there is no moral hazard problem at the originating bank. The bank does not have an incentive to grant unprofitable loans because nobody will be willing to insure the risks from such loans. Hence, CRT is desirable from a welfare perspective.

If, however, loans are granted on the basis of privately observable information, the transfer of credit risk is hampered by problems of asymmetric information. If credit insurers cannot observe a loan's quality, banks have an incentive to grant unprofitable loans and to transfer the risks from these loans to the insurers. This is anticipated by the credit insurers who will demand a lemons premium for credit risk transfer. CRT generally still improves the access to finance for risky, but profitable borrowers, but it also improves the access to finance for risky, but profitable borrowers, but it also improves the access to finance for unprofitable borrowers. As a result, the aggregate risk in the economy increases. Note that, in our model, the overall welfare effect of CRT is positive even with private information. The reason is that the positive welfare effects from a better access to finance for profitable borrowers overcompensate the welfare losses from financing projects with negative net present values (NPV).<sup>5</sup>

<sup>&</sup>lt;sup>5</sup>The paper by Parlour and Plantin (2008) yields similar findings regarding the incentive effects of

We show further that competition generally reinforces the positive effects of credit risk transfer. The higher competition in the banking sector, the better is the access to credit. However, with private information, an increase in competition may reduce welfare when the loan market for profitable loans is saturated. Then, a further increase in competition only improves the access to credit for unprofitable borrowers. In the limit, this completely offsets the welfare gains from credit risk transfer. This finding coincides nicely with the observations from the current crisis. During the late years of the credit boom preceding the crisis, most of the newly extended loans seem to have been of relatively poor quality; at the same time, these years saw an increase in competition through the market entry of new financial institutions (see Dell'Ariccia, Igan, and Laeven, 2008).

Finally, we show that the CRT market breaks down if competition is very low. The reason is that low competition goes along with large risk-bearing capacities, which implies that banks keep most profitable loans on their own balance sheets, such that the average quality of loans to be insured is low and insurers are no longer willing to insure such loans. This may explain why CRT markets developed in an environment of banking deregulation and increasing competition.

To sum up, our paper illustrates two important points. First, it describes how CRT may lead to a moral hazard problem in the origination of loans. When information is private, CRT induces banks to knowingly extend negative NPV loans, leading to an increase in aggregate risk as seen in the recent crisis. Second, and more importantly, it shows that the welfare consequences of CRT depend on the degree of competition in the banking sector and on the type of information on which loans are based. The introduction of CRT markets generally leads to an increase in welfare because it improves the access to finance for profitable borrowers. However, under private information, higher competition leads to an expansion of negative NPV loans, which in the limit offsets the welfare gains from CRT completely.

The paper proceeds as follows. Section 2 presents the basic model setup. Section 3 describes the equilibrium of the model in the absence of credit risk transfer. Section 4 analyzes the functioning of CRT markets when loans are granted on the basis of public and private information, respectively. Section 6 concludes. Proofs are in the Appendix.

CRT, although it deals with monitoring rather than screening. Interestingly, their results on welfare are contrary to ours. We will discuss the reasons behind this difference at the end of the paper.

# 2 Model Setup

Our model has several important ingredients. First, borrowers are heterogenous; they differ in their creditworthiness. Second, competition among banks is imperfect; specifically, we use a model of Salop competition. Third, banks have limited risk-bearing capacities, for example due to regulation. These features allow us to model lemons problems in CRT markets and the impact of competition on banks' risk-bearing capacities and hence on the potential of CRT to improve borrowers' access to credit. Later, we will distinguish between loans based on public or private information. This will be important in the discussion of CRT markets because the type of information will substantially affect the functioning of these markets.

**Entrepreneurs.** Consider an economy with a continuum of entrepreneurs. Each entrepreneur has access to a project that requires an investment of one unit of money. In order to finance their projects, entrepreneurs must take up a bank loan. Projects have one of three qualities, they are either good (G), medium (M), or bad (B).<sup>6</sup> Entrepreneurs with good (medium, bad) projects are called good (medium, bad) entrepreneurs. Projects have a positive return of Y with probability  $p_i, i \in \{G, M, B\}$ ; otherwise they fail and return nothing. Projects of the same type are perfectly correlated. A share  $q_G$  of all projects is good. Good projects succeed with probability  $p_G = 1$ . They have a positive net present value,  $p_G Y - r = Y - r > 0$ , where r is the opportunity cost of one unit of money. A share  $q_M$  of all projects is medium. Medium projects succeed with probability  $p_M < 1$ , but they also have a positive net present value,  $p_M Y - r > 0$ . A share  $q_B = 1 - q_G - q_M$ of all projects is bad. Bad projects succeed with probability  $p_B < p_M < 1$ , and their net present value is negative,  $p_B Y - r < 0$ . As a result, there are two kinds of projects that are desirable from a social perspective: the good projects, which are safe, and the medium project, which are risky. The third class of projects is so risky that they are undesirable from a social perspective.

<sup>&</sup>lt;sup>6</sup>These entrepreneurs can also be interpreted as borrowing households with different risk profiles.



The figure shows a Salop circle with n = 7 banks that are distributed equidistantly on the Salop circle. The three types of entrepreneurs, continuously located on the Salop circle, all have equal shares,  $q_G = q_M = q_B = 1/3$ . Dark gray stands for good entrepreneurs, medium gray for medium ones, and light gray for bad ones.

Banking Market Structure. Banks compete for loans à la Salop (1979).<sup>7</sup> They announce loan rates  $R_G$ ,  $R_M$ , and  $R_B$  for entrepreneurs with good, medium, and bad projects.<sup>8</sup> The entrepreneurs are uniformly distributed on a circle of length L, which is normalized to 1 (see Figure 1). Hence, the aggregate volume of potential projects is L = 1, and  $q_i L = q_i$  is the aggregate volume of potential projects of type i. In order to obtain a loan, an entrepreneur must travel to the bank, incurring transportation costs tper unit of distance.<sup>9</sup> When choosing a bank, the entrepreneurs take into account both transportation costs and interest rates. The banks are distributed equidistantly on the Salop circle. There is no equity; the only source of refinancing is deposits, which are offered at a gross interest rate r, including the repayment of the principal. Deposits are fully insured, and the costs of deposit insurance are normalized to zero. If a bank's liabilities exceed the returns from its loans, it defaults. We assume that the number of banks, n, is fixed. Below, we will consider free entry in the banking sector.

<sup>&</sup>lt;sup>7</sup>The Salop model has frequently been used to model loan market competition in the banking sector. See Freixas and Rochet (1997) for an overview. Alternative models of price competition, e. g. monopolistic competition as in Monti (1972), Klein (1973), and Shubik and Levitan (1980), yield similar results.

<sup>&</sup>lt;sup>8</sup>The Salop model assumes that banks do not price discriminate among entrepreneurs at different locations. In Section 5, we discuss a model with observable locations.

<sup>&</sup>lt;sup>9</sup>See Degryse and Ongena (2005) for empirical evidence that transportation costs are important in loan markets.

- t = 0: Banks announce loan rates, depending on the borrowers' qualities
- Borrowers choose a bank and invest
- Banks enter the market for credit risk transfer (if applicable)
- t = 1: Borrowers repay their loans if they are successful, otherwise they fail. If a loan to a failing borrower has been insured, the credit insurer repays the loan. Banks repay deposits if they can, otherwise they fail.

**Screening Technology.** Banks have access to a screening technology to find out the quality of an entrepreneur's project. The technology produces a noiseless signal. Later in the paper, we will distinguish between two kinds of screening technologies, based on either public or private information.

**Banks' Probability of Default.** Finally, we assume that banks are regulated to have a probability of default below some level  $\alpha$ .<sup>10</sup> As we will see later, this assumption constrains the banks' risk-bearing capacities and hence firms' access to credit, yielding a rationale for credit risk transfer.<sup>11</sup>

The time structure of the game is given in Figure 2.

# 3 No Credit Risk Transfer

We will now show that the described setup with no possibility of transferring risks to other parties leads to a situation where banks are constrained in their lending due to their restricted risk-bearing capacities. In particular, loans to medium entrepreneurs will

<sup>&</sup>lt;sup>10</sup>This could be achieved by imposing capital requirements on the basis of a bank's value at risk at a confidence level  $1 - \alpha$ .

<sup>&</sup>lt;sup>11</sup>Pennacchi (1988) was the first to motivate credit transfer by regulation. Alternatively, the desire for CRT could arise from bankruptcy costs (as in Wagner and Marsh, 2006) or from bankers' risk aversion (as in Morrison, 2005).

be below their optimal level. Interestingly, fiercer competition (through bank entry) is shown to tighten banks' lending constraints.

Access to credit. The loan volume granted by a single bank to borrowers of type i is denoted by  $l_i$ ; the aggregate volume of loans of type i is denoted by  $L_i = n l_i$ . Bad projects have a negative net present value, hence bad entrepreneurs do not have access to loans in equilibrium, i. e.  $l_B = 0$ . A bank's probability of default is determined by its loan volumes and loan rates. With probability  $p_M$ , both good and medium loans repay, and the bank's profit is  $(R_G - r) l_G + (R_M - r) l_M > 0$ . With probability  $1 - p_M$ , only the good loans repay, and the profit is  $(R_G - r) l_G - r l_M$ . If this term is (weakly) positive, then the bank's probability of default is zero. If this term is negative, then the bank's default probability  $\alpha$  has to be smaller than  $1 - p_M$ , implying that

$$(R_G - r) l_G - r l_M \ge 0.$$
 (1)

In the following, we assume that the regulation has an effect, such that condition (1) binds in equilibrium. We restrict our attention to situations where the good loan market is covered completely, such that banks compete for loans at least in this loan segment. This will always be true when banking markets are sufficiently competitive (e.g., n is sufficiently large or t is sufficiently small). We can then calculate  $l_G$  by deriving the distance  $x_G$  between a bank and a good borrower who is just indifferent between a loan from the bank at a loan rate  $R_G$  and a loan from the neighboring bank at a loan rate  $R'_G$ .

$$(Y - R_G) - t x_G = (Y - R'_G) - t (1/n - x_G).$$
<sup>(2)</sup>

Solving for  $x_G$  and considering that  $l_G = 2 q_G x_G$ , we get

$$l_G = q_G \left(\frac{1}{n} + \frac{R'_G - R_G}{t}\right). \tag{3}$$

Condition (1) implies that banks cannot grant as many medium loans as they would like to in the absence of regulation; the access to credit is constrained for medium entrepreneurs due to limited risk-bearing capacities of banks. Hence, the market for medium loans is not covered and banks enjoy local monopolies in the segment for medium loans, as depicted in Figure 3.



The figure shows a typical market share of a bank. Note that the good loan segment is covered completely, but not the medium loan segment. Banks do not grant any bad loans.

We can then calculate the distance of a medium borrower  $(x_M)$  who is just indifferent between a loan from a bank at a loan rate  $R_M$  and no loan at all,

$$p_M (Y - R_M) - t x_M = 0. (4)$$

Solving for  $x_M$  and considering that  $l_M = 2 q_M x_M$ , we get

$$l_M = 2 \, p_M \, q_M \, \frac{Y - R_M}{t}. \tag{5}$$

Banks maximize their expected profits,

$$\Pi = (R_G - r) \, l_G + (p_M \, R_M - r) \, l_M, \tag{6}$$

subject to condition (1). This maximization yields  $R_G$ ,  $R_M$  and  $\lambda$ , the shadow price of condition (1). Proposition 1a characterizes the described equilibrium.<sup>12</sup>

**Proposition 1a (Equilibrium without CRT)** There is an equilibrium in which the market for good loans is covered completely, the market for medium loans is not covered completely, and the shadow price  $\lambda$  of condition (1) is strictly positive. This equilibrium obtains if

•  $q_M/q_G > (Y-r)/(2r)$ ,

 $<sup>^{12}\</sup>mathrm{The}$  proofs of all propositions can be found in the Appendix.

- $n \ge t/(Y-r)$ , and
- $n > t \sqrt{q_G} / \sqrt{q_M r (p_M Y r)}.$

The first two conditions guarantee that the good market is covered, but not the medium market. The third condition implies that the shadow price  $\lambda$  is strictly positive.

Some socially beneficial, but risky projects (of type M) are not carried out because banks have to avoid default to satisfy regulatory constraints. We will see later that this restriction can be eased by introducing a market for credit risk transfer (Section 4).

**Competition.** According to condition (1), a bank's risk-bearing capacity is determined by its profits from the good loan segment. These depend on the intensity of competition in the banking sector. This leads to the interesting result that the access to credit for medium firms is *reduced* by fiercer competition. Proposition 1b summarizes the effects of competition on firms' access to credit.

#### **Proposition 1b (Competition)** *Higher competition (higher n)*

- leaves the aggregate amount of good loans unaffected,  $dL_G/dn = 0$ ,
- lowers the aggregate amount of medium loans,  $dL_M/dn < 0$ ,
- increases the shadow price of condition (1),  $d\lambda/dn > 0$ .

Surprisingly, more banks lead to a lower market penetration for medium loans (see the second chart of Figure 4, which is based on a numerical example). The reason is the following. When competition intensifies, the banks' margins in the good loan segment shrink due to decreasing loan rates for good loans,  $R_G$  (see the first chart of Figure 4). These margins determine how aggressive banks are in the medium loan segment because banks have to comply with condition (1). The lower the profits in the good loan segment, the lower are the banks' buffers against default, and the fewer medium loans they are willing to grant. Hence, loan rates  $R_M$  increase, and the market penetration in the medium loan segment declines. In other words, the underprovision of loans in the medium segment (i.e. profitable, but risky loans) is most severe when there is fierce competition

Figure 4: Comparative Statics with Respect to n



Colors are the same as in Figure 1: Light gray stands for bad borrowers, medium gray for medium borrowers, and dark gray for good borrowers. Parameters for the numerical example are  $q_G = q_M = q_B = 1/3$ ,  $p_G = 1$ ,  $p_M = 2/3$ ,  $p_B = 1/3$ , Y = 2, r = 1, and t = 2. The larger the number of banks, the higher the competition for good borrowers, which shows up in a lower loan rate  $R_G$ . The volume of good loans  $L_G$  is constant because the whole market is always covered. As *n* increases, banks' buffers decrease and banks become less aggressive in the medium loan segment and raise  $R_M$ . Hence, the aggregate volume of medium loans  $L_M$  decreases. The shadow price of condition (1) increases in *n* because banks are more constrained in their lending to medium borrowers. The rents of good entrepreneurs ( $W_G$ ) increase in *n*, due to the decrease in loan rates  $R_G$ . The rents of medium entrepreneurs ( $W_M$ ) decrease in *n* due to higher loan rates  $R_M$  and a lower loan volume  $L_M$ . Finally, banks' profits (*n* II) decrease. Aggregate welfare (W) is non-monotonic. For this numerical example, it reaches its maximum at  $n^* = 6$ .

in the banking sector.  $\lambda$  is a measure of how much a bank suffers from having to adhere to condition (1). A higher  $\lambda$  implies a higher marginal profit of the banks when condition (1) is relaxed. The third chart of Figure 4 shows that a higher number of banks leads to an increase in  $\lambda$ , reflecting the tighter constraints on banks' lending in the medium segment.

In Proposition 1b, the degree of competition is identified with the number of banks n. However, the intensity of competition is also affected by transportation costs t. When transportation costs decrease, "shopping around" for loans becomes easier. As can be seen from the proof of Proposition 1b in the Appendix, a decline in t has similar consequences as an increase in n. In particular, the volume of medium loans decreases, and the shadow price  $\lambda$  increases. Welfare. Within this setting, utilitarian welfare consists of four parts: aggregate rents of the three types of entrepreneurs, and aggregate profits of banks. Bad entrepreneurs do not receive any loans, hence their rents are zero. Welfare for good and medium entrepreneurs is equal to the rents of entrepreneurs who receive a loan. Interestingly, and in contrast to the ordinary Salop model, aggregate welfare is *not* strictly increasing in the number of banks n, although there are no entry costs.

**Proposition 1c (Welfare)** If  $4 p_M Y > 5 r$ , the welfare function is non-monotonic in the number of banks n and reaches a maximum at

$$n^* = t \sqrt{\frac{3 q_G}{q_M r \left(4 p_M Y - 5 r\right)}}.$$
(7)

For large n, welfare converges to  $q_G(Y-r)$ , the aggregate NPV of good projects.

The reason behind this result is the banks' lending constraints. Increasing competition decreases banks' ability to lend to (profitable) medium entrepreneurs.<sup>13</sup> Since the banks' margins in the good loan segment converge to zero, banks cannot grant any medium loans in the limit. Therefore, for  $n \to \infty$ , welfare converges to the aggregate NPV of good projects. The welfare loss of excessive competition in the banking sector can be arbitrarily large, for example if medium projects have a large net present value,  $p_M Y \gg r$ .

Summing up, there is a welfare-optimal degree of competition. Welfare reaches a maximum when (7) holds. As before, competition can be identified just as well with the size of transportation costs t.

**Market Entry.** In order to analyze the effects of free entry in the banking sector, let us assume that there are fixed entry costs f. Banks will then enter until  $\Pi = f$ . Hence, the number of entering banks is

$$n = \sqrt{q_G t \frac{p_M Y + \sqrt{p_M^2 Y^2 - 2 f t/q_M}}{2 f r}}.$$
(8)

Here, the fixed entry costs f deter banks from entering, dn/df < 0, as is always the case in Salop models with free entry. The reason is that the higher fixed costs can only be

<sup>&</sup>lt;sup>13</sup>In the absence of medium entrepreneurs,  $q_M$  would be zero, and  $n^*$  would converge to  $\infty$ , which means that welfare would be a strictly increasing function.

earned in equilibrium if there is less competition and, hence, higher margins. As a result, all other endogenous variables  $(l_G, l_M, R_G, R_M, \text{ and } \lambda)$  respond to a change in fixed costs f. The comparative statics with respect to f are analogous to those with respect to n(but with the opposite sign). An increase in f leaves the aggregate amount of good loans unchanged, but it increases the market penetration in the medium loan segment. The shadow price of condition (1) decreases in f because banks are less constrained in their lending to medium borrowers. As a result, higher entry costs lead to larger banks for two reasons. First, fewer banks enter, increasing the loan volume in the good market segment. Second, banks expand their medium loans due to higher buffers in the good segment.

As in the traditional Salop model, there is excessive entry in this model. The reason is that firms do not take into account the negative externality that their entry has on the other firms' profits. In this model, there is a second externality that exacerbates the excessive entry problem. Banks do not take into account the negative externality that their entry has on their competitors' risk-bearing capacities, and hence on the credit availability for medium entrepreneurs.

# 4 Credit Risk Transfer

We now allow banks to transfer risks from their balance sheets to other investors. For simplicity, we model credit risk transfer as an insurance contract with outside investors. The possibility of trading credit risk relaxes condition (1). Hence, CRT may improve the access to credit for medium entrepreneurs. The higher the shadow price of condition (1), the higher the benefits of banks from transferring their credit risks. We will see, however, that the functioning of CRT markets depends crucially on the type of information underlying the banks' loans. We will distinguish between two types of information: public information and private information.

### 4.1 Model Setup

**Insurers.** Outside of the banking system, there is a continuum of risk neutral investors who are willing to insure the banks against credit default at a fair premium.<sup>14</sup> We assume

<sup>&</sup>lt;sup>14</sup>As an alternative, one could assume that there are several economies, each of which contains a Salop circle. Then banks in different economies can share their credit risk. For a large number of economies,

that the market for CRT is anonymous, such that the amount of transferred credit risk is unobservable by the insurers. This seems to be reasonable given the opaqueness and complexity of CRT markets.

**Public vs. Private Information.** We assume that the information produced by a bank in the screening process is either publicly or privately observable by the bank. If the information is publicly observable, it can also be observed by potential insurers. If the screening information is privately observable, it is not observable by potential insurers. The distinction between public and private information is related to that between *hard* and *soft* information, popularized by Stein (2002) (see also Petersen, 2004). For example, a credit rating that is produced by a standardized statistical rating system on the basis of balance sheet information is hard information and is publicly observable by other parties. In contrast, the personal impression of the loan officer during the loan interview is soft information and cannot be publicly observed.

In the following, we will discuss the properties of an equilibrium with credit risk transfer under public and private information. In each case, we will start by analyzing the effect of CRT, holding the number of banks fixed; we will then allow for free market entry.

### 4.2 Public Information

In this section, we assume that the banks' screening technologies produce publicly observable information about the entrepreneurs. Hence, if a bank wants to transfer its credit risk, it can communicate the quality of the underlying loans to the insurer. As a consequence, only medium loans are insured. Good loans do not entail any risk, so there is no benefit from credit insurance; bad loans are not granted in the first place.

An insurance contract will allow the bank to turn a risky loan into a safe payment. If a bank insures a medium loan, it pays a premium  $\pi_M R_M$  to the insurer at date t = 0, independent of whether the loan eventually fails or not.<sup>15</sup> Risk neutral competitive

the effect would be the same as with risk neutral insurers. Potentially, banks could perfectly diversify their risk at zero cost.

<sup>&</sup>lt;sup>15</sup>Since banks choose not to default in equilibrium, it is irrelevant whether they settle the insurance premia in t = 0 or t = 1.

insurers break even if  $\pi_M - (1 - p_M)/r = 0$ , hence  $\pi_M = (1 - p_M)/r$ . The bank has to refinance the premium by taking up more deposits, implying an additional repayment to depositors at date t = 1 of  $\pi_M r R_M = (1 - p_M) R_M$ . At date t = 1, the insurer pays  $R_M$ to the bank if a medium loan fails; if the loan does not fail, the bank receives  $R_M$  from the debtor. Therefore, the bank receives  $R_M (1 - (1 - p_M)) = p_M R_M$  with certainty. Hence, the bank has used insurance contracts to turn a risky loan with an expected payment of  $p_M R_M$  into a safe payment of  $p_M R_M$ .

Due to the possibility of unloading credit risk, condition (1) no longer applies. The bank can grant more medium loans. Hence, the introduction of credit risk transfer improves firms' access to finance in the case of public information. We start by considering the effects of the introduction of a CRT market, holding the number of banks constant. Two kinds of equilibria may result. First, the market penetration for medium loans improves, but banks still do not cover the entire circle. Second, the market for medium loans may be covered completely, like the market for good loans. In the second case, the analysis boils down to that of a standard Salop model with two separate loan markets. Let us therefore concentrate on the first case.<sup>16</sup>

The bank again maximizes its expected profits,

$$\Pi = (R_G - r) \, l_G + (p_M \, R_M - r) \, l_M, \tag{9}$$

where  $p_M R_M l_M$  is now a safe payment because medium loans are covered by credit insurance. Condition (1) is no longer binding, implying that  $\lambda$  is equal to 0. Hence, profits are higher than in the absence of a CRT market. The market for good loans is again covered completely. The market for medium loans is not covered, but the penetration is larger than in the situation with CRT. The following proposition summarizes the effects of the introduction of a CRT market with public information. See Figure 5 for a graphical illustration.

**Proposition 2a (Credit Risk Transfer with Public Information)** With public information, the introduction of credit risk transfer

• leaves the aggregate amount of good loans unaffected,

<sup>&</sup>lt;sup>16</sup>The second case is discussed in the Appendix.

- increases the aggregate amount of medium loans,
- reduces the shadow price of condition (1) to zero,  $\lambda = 0$ ,
- increases banks' expected profits.

**Competition.** We can now analyze how competition affects banks' behavior in the presence of CRT with public information. In the presence of credit risk transfer, as long as the medium loan segment is not covered completely, banks act as monopolists on the medium loan market. Hence the medium loan volume of a single bank does *not* depend on the number of banks. As a direct consequence, the penetration in the medium segment is proportional to *n*. This result stands in contrast to Proposition 1b. A higher number of banks and the resulting increase in competition *improve* the access to credit for medium entrepreneurs when there is a functioning CRT market, whereas they worsen the access to loans in the absence of credit risk transfer (see Figure 5). Proposition 2b summarizes these results.

**Proposition 2b (Competition)** In the presence of credit risk transfer with public information, higher competition (higher n)

- leaves the aggregate amount of good loans unaffected,  $dL_G/dn = 0$ ,
- increases the aggregate amount of medium loans until the market is covered completely,  $dL_M/dn \ge 0$ .

Welfare. The introduction of CRT with public information leads to a Pareto improvement. We have seen already that banks' expected profits increase (see Proposition 2a). Moreover, medium borrowers benefit for two reasons. First, the borrowers who had access to credit even in the absence of CRT benefit from lower loan rates. Second, other medium borrowers profit from gaining access to credit. For  $n \to \infty$ , all good and medium entrepreneurs get loans and carry out projects, and transportation costs converge to zero. Welfare converges to the aggregate NPV of good and medium projects.

Figure 5: Credit Risk Transfer with Public Information



Colors are the same as in Figure 1: Light gray stands for bad borrowers, medium gray for medium borrowers, and dark gray for good borrowers. The solid lines refer to a situation with public-information CRT. In the left picture, the dashed line shows the loan volume in the medium loan segment in the absence of CRT (for good and bad loans the volumes are unchanged by the introduction of CRT). We see that CRT leads to a higher market penetration in the medium loan segment. Moreover, the market penetration for medium loans now weakly increases in n; for large enough n, the market is covered completely. In the right picture, the solid line refers to aggregate welfare in the presence of CRT, the dashed line to aggregate welfare in the absence of CRT (as in Figure 4). The welfare function has a kink at the point where the medium loan market is saturated.

**Proposition 2c (Welfare)** The introduction of credit risk transfer with public information increases aggregate welfare. The welfare function is strictly increasing in the number of banks n and converges to  $q_G(Y - r) + q_M(p_M Y - r)$ , the aggregate NPV of good and medium projects.

Note that this welfare result hinges on the fact that the credit risk transfer leads to an efficient transfer of (macroeconomic) risk to insurers who are better able to bear the risks than banks. However, it is not necessary to assume that insurers are strictly risk neutral. If they were risk averse, they would demand a risk premium for taking the (macroeconomic) risk. In equilibrium, this would result in higher loan rates for medium entrepreneurs and smaller loan volumes. Nevertheless, welfare would still increase due to the introduction of CRT markets. Insurers would earn a non-negative rent (otherwise, they would not participate), medium entrepreneurs would benefit from increased access to loans and from reduced interest rates, and banks would benefit from increased profits. Welfare gains would, however, depend negatively on the degree of insurers' risk aversion. If the insurers' risk aversion is too large, there will be no CRT. The maximum premium that banks are willing to pay will depend on the shadow price  $\lambda$ .

Market Entry. Consider again a situation in which, in the long run, banks enter until their expected profits  $\Pi$  equal some fixed entry costs f. On the basis of Propositions 2a and 2b, we can derive the long-run effects of the introduction of a CRT market with public information, allowing for market entry. The number of banks in equilibrium is

$$n = t \sqrt{\frac{q_G}{f t - q_M (p_M Y - r)^2/2}}.$$
(10)

Since the introduction of CRT increases expected profits in the short run (cf. Proposition 2a), it attracts more banks in the long run. This implies that there is more competition for good loans. As a consequence, the loan rate  $R_G$  decreases. The aggregate volume of good loans cannot change because the market is already completely covered. For medium loans, banks have local monopolies. As argued above, the volume of medium loans  $l_M$  of a single bank does not depend on n. Hence, the market penetration for medium loan is proportional to n; it increases as more banks enter the market. Hence, the effects of CRT on medium loans are reinforced in the long run through market entry.

### 4.3 Private Information

When loans are based on private information, banks cannot credibly communicate the quality of a loan underlying an insurance contract. Therefore, the transfer of credit risk becomes more difficult. The asymmetric information about loan qualities leads to a moral hazard problem. Banks knowingly grant bad loans only to resell them to the insurers.<sup>17</sup> This is anticipated by the insurers who demand a lemons premium. Under some circumstances, the market for credit risk transfer even breaks down completely.

In equilibrium, the insurers anticipate the underlying credit risk and set their insurance premia accordingly. When deciding whether to grant a loan and whether to insure the risk from that loan, banks take insurance premia, and hence the price of CRT, as given. Let  $\beta$ denote the insurers' anticipated probability that an underlying loan has *medium* quality. The loan is then expected to be *bad* with probability  $1 - \beta$ . Good loans are not risky, hence banks never insure such loans. Therefore, the insurer expects an underlying firm of an insurance contract to be successful with an average probability of  $\bar{p} \equiv \beta p_M + (1-\beta) p_B$ .

<sup>&</sup>lt;sup>17</sup>This is different from the incentive problem analyzed by Pennacchi (1988), Gorton and Pennacchi (1995), and Chiesa (2008), where banks have suboptimal incentives to monitor loans transferred to other parties. Here, banks know that a loan is of low quality, but they still decide to grant it.

The insurer anticipates a firm to default with probability  $1 - \bar{p}$ , entailing a payment to the bank.<sup>18</sup>

As above, with an insurance contract, a bank can turn a risky yield from a loan into a safe payment. Let us give an example. A medium loan allows the bank to claim  $R_M$  from the entrepreneur. In order to completely insure this loan, the bank must buy a volume  $R_M$  of credit protection. The premium will be  $R_M \pi = R_M (1-\bar{p})/r$ . Taking into account the loan, the insurance contract and the costs of refinancing the insurance premium, the bank will then receive a safe payment of  $\bar{p} R_M$  at date t = 1. Hence, a bank makes a negative expected profit from insuring a medium loan  $(\bar{p} R_M < p_M R_M)$ ; in contrast, the bank makes a positive expected profit from insuring a bad loan. In both cases, selling risk has the positive effect that condition (1) is relaxed. The bank can thus expand and grant more loans. As a consequence, the bank never resells the entire risk within its portfolio. Condition (1) must bind in equilibrium. If it did not bind, the bank could increase profits by insuring fewer medium risks. Also, a bank never grants a bad loan and keeps it in its balance sheet. All bad loans are resold in equilibrium.

As in (5), a single bank's volume of medium loans is  $l_M = 2 p_M q_M (Y - R_M)/t$ , and the volume of bad loans is  $l_B = 2 p_B q_B (Y - R_B)/t$ , accordingly. Let  $\kappa$  be the fraction of medium loans that a bank insures, and  $1 - \kappa$  the fraction of medium loans that remain in the bank's balance sheet. A bank's total loan volume (and hence the balance sheet total) is  $l_G + l_M + l_B$ , and refinancing costs are  $(l_G + l_M + l_B)r$ . In the best possible case (with probability  $p_M$ ), all loans are repaid; the bank gets  $R_G l_G$  from the good loans,  $\bar{p} (l_B R_B + \kappa l_M R_M)$  from the insured bad and medium loans (net of the payments to insurers), and  $(1 - \kappa) l_M R_M$  from uninsured medium loans. In the worst possible case (with probability  $1 - p_M$ ), medium entrepreneurs do not repay, all other payments are identical. Hence, condition (1) is modified to

$$(R_G - r) l_G + \bar{p} (l_B R_B + \kappa l_M R_M) - (l_M + l_B) r \ge 0.$$
(11)

Obviously, an increase in the sale of credit risk (i.e. an increase in  $\kappa$ ) relaxes (11).

<sup>&</sup>lt;sup>18</sup>Note that, with private information, there can be multiple equilibria, depending on whether insurers expect a high or a low average success probability. Both types of beliefs can be justified in equilibrium. In the following, we will focus on the Pareto-efficient equilibrium with larger  $\bar{p}$  and lower interest rates. A discussion of the other equilibrium can be found in the proof of Proposition 3a in the Appendix.

Because this condition is binding, it implicitly defines  $\kappa$ . Banks choose loan rates in order to maximize their expected profits,

$$\Pi = [R_G - r] l_G + [(\bar{p} \kappa + p_M (1 - \kappa)) R_M - r] l_M + [\bar{p} R_B - r] l_B, \qquad (12)$$

subject to condition (11) and taking  $\beta$  (and hence  $\bar{p}$ ) as given.  $\bar{p} = \beta p_M + (1 - \beta) p_B$  can be determined by using Bayes' law. The probability that an insured loan has medium quality is

$$\beta = \frac{\kappa \, l_M}{\kappa \, l_M + l_B}.\tag{13}$$

Substituting the binding constraint (11) in the bank's profit function (12) and considering the first-order conditions with respect to  $R_M$  and  $R_B$  yields the interesting result that

$$R_M = R_B = \frac{\bar{p}\,Y + r}{2\,\bar{p}}.\tag{14}$$

Although bad loans are riskier than medium ones, and hence have a lower NPV, the loan rate for both types of entrepreneurs is identical. The intuition goes as follows. In each market, the loan rate is determined by the expected payments that the last infinitesimal loan (the marginal loan) earns. However, given that the bank is constrained and inequality (11) binds, the bank must insure the marginal loan. Consequently, the probability of success becomes irrelevant. In equilibrium, the bank offers the same loan rates for medium and bad loans.

Proposition 3a summarizes the effects of the introduction of a CRT market with private information.

**Proposition 3a (Credit Risk Transfer with Private Information)** With private information, if the market for credit risk transfer does not break down, its introduction

- leaves the aggregate amount of good loans unaffected,
- increases the aggregate amount of medium loans, but less than with public information,
- increases the aggregate amount of bad loans,
- reduces the shadow price of condition (1), but not to zero,  $\lambda > 0$ ,

• increases banks' expected profits, but less than with public information.

The directions of all effects are the same as for CRT with public information. However, because banks also insure bad loans, they have to pay a lemons premium. As a result, the loan rate for medium loans  $R_M$  is higher than under CRT with public information, and the volume  $L_M$  is lower. Because  $R_B = R_M$ , the volume of bad loans is positive. Hence, the introduction of CRT improves the access to credit for medium entrepreneurs less than with public information. In addition, CRT improves the access to credit for bad entrepreneurs with negative net present values. Given that the reason for CRT, and hence ultimately the reason for the existence of negative NPV loans, is condition (1), this result is in line with the findings of Keys, Mukherjee, Seru, and Vig (2009b). The quality of loans may be worse in a more regulated banking system.

**Competition.** Consider now an increase in the number of banks. Due to stronger competition for good loans,  $R_G$  decreases, but  $L_G$  remains constant. As a consequence, the profits from the good loan segment that banks can put at risk in the other loan segments decrease. We have argued above that banks keep only medium loans in their balance sheets; all bad loans are insured. When buffers decrease, banks can keep relatively fewer medium loans in their own balance sheets. The average quality of insured loans improves, and the lemons premium drops. Since the lemons premium is a component of the price of credit risk transfer, the price of CRT drops, and banks can grant more loans. The loan rates  $R_M$  and  $R_B$  drop.

The opposite happens when the number of banks goes down. Then, CRT becomes more expensive because the banks prefer to keep more medium loans on their own balance sheets. At some point, for small enough n, the market for CRT breaks down completely. Analogously, the market will break down for large enough transportation costs t. These results are summarized in the following proposition.

**Proposition 3b (Competition)** In the presence of credit risk transfer with private information, if the market does not break down, higher competition (higher n)

• leaves the aggregate amount of good loans unaffected,  $dL_G/dn = 0$ ,

- increases the aggregate amount of medium loans until the market is covered completely,  $dL_M/dn \ge 0$ ,
- increases the aggregate amount of bad loans,  $dL_B/dn > 0$ ,
- increases the fraction of medium loans that is insured,  $d\kappa/dn > 0$ .

For some  $\bar{n} > 0$ , the market for credit risk transfer breaks down for  $n \leq \bar{n}$ .

The last part of Proposition 3b offers a new explanation for the evolution of CRT markets before the crisis. In the interpretation of our model, the strong increase in the use of CRT instruments before the financial crisis may have been due to the intensification of competition in the banking sector. When competition in the banking sector was low, banks' balance sheets were healthy enough to absorb large amounts of risk. Consequently, CRT was not necessary, and banks granted loans with an efficient risk-return structure. When competition increased, e.g., due to financial deregulation such as the abolishment of the Glass-Steagall act in 1999, or the branching deregulation initiated in the Riegle-Neal act of 1994, capital buffers of banks shrank and banks needed to shed more risk. The market for CRT became active, and banks had an incentive to grant negative-NPV high-risk loans, just in order to resell them. Hence, financial deregulation may have been one reason for financial instability. The proposition is also in line with Vickery (2007) who shows that savings banks (operating in a less competitive environment) retain mortgages on their own balance sheet and originate loans with low levels of risk, whereas finance companies (operating in a more competitive environment) choose an originateand-distribute strategy and grant riskier loans. Note again that fiercer competition can also be caused by a reduction in "transportation costs." In fact, it seems that in the years before the current crises, "shopping around" for loans has become much easier in retail banking in the U.S., not least due to the advent of the internet and the resulting increase in price transparency.<sup>19</sup>

The results of Proposition 3b are illustrated in Figure 6. There is a critical n (in the numerical example  $n \approx 14$ ) below which the market for CRT breaks down. Only for higher n, a market for CRT with private information can be maintained. The reason is the following. For low levels of competition, banks earn high profits on good loans,

<sup>&</sup>lt;sup>19</sup>We thank an anonymous referee for bringing this point to our attention.



Figure 6: Credit Risk Transfer with Private Information

Colors are the same as in Figure 1: Light gray stands for bad borrowers, medium gray for medium borrowers, and dark gray for good borrowers. The first chart of the figure shows aggregate loan volumes. The solid lines again refer to a situation with CRT. The dashed lines show the loan volumes in the medium and bad loan segments in the absence of CRT (for good loans the volume is unchanged by the introduction of CRT). If the market for CRT does not break down, there is an increase in both medium and bad loans due to CRT; the market penetration for medium and bad loans increases in n. The second chart shows the fraction of transferred medium loans  $\kappa$ , together with the fractions of transferred good loans (which is always zero) and bad loans (which is equal to 1 as long as the market for CRT is active). The higher n, the fewer medium loans can banks keep in their books. Notice the discontinuity of most curves at  $n \approx 14$ , which is where the CRT market starts to exist. Most curves also exhibit a kink at  $n \approx 19$ , at which point the market for mediums loans is saturated. The third chart shows welfare. Welfare without CRT is dashed black, welfare with public-information CRT is solid black, and welfare with private-information CRT is gray. The gray welfare function increases as long as the medium loan market expands. Above the kink, only the bad loan market expands, and welfare decreases.

hence they need to insure fewer medium loans; as a consequence, the price for CRT is high, leading to a complete market breakdown because the lemons problem becomes too severe. All variables are then equal to those in the absence of CRT (cf. Figure 4). For higher n, the CRT market is active. As n becomes larger, the penetration of the medium loan segment increases, until at some point ( $n \approx 19$  in the figure) the medium market is covered completely. The bad loan segment keeps growing. As a consequence, the lemons premium for CRT increases, and the growth of bad loans is (slightly) smaller. Consequently, competition has the beneficial effect of increasing the market penetration for medium loans, but it inflates the volume of bad loans at the same time.

When the medium market is covered completely, a further increase in competition improves only the access to credit for negative NPV firms. Given that  $l_M = 2 q_M p_M (Y - R_M)/t$ , the medium market is saturated when  $R_M = Y - t/(2 n p_M)$ . Since  $R_B = R_M$ , the volume of bad loans will be  $L_B = q_B p_B/p_M < q_B$  at this point, hence the bad market is not saturated. The reason is the following. For a given loan rate and a given distance to the closest bank, expected profits of medium entrepreneurs are higher than those of bad entrepreneurs, due to the higher success probability of medium projects. Consequently, the medium loan market is always penetrated more than the bad market. When the medium market is saturated, fiercer competition cannot lead to a further increase in  $L_M$ . Consequently, the price of credit insurance increases, which makes bad loans  $L_B$  grow more slowly. This is also visible in Figure 6.

The maximum volume of bad loans is bounded not only by the size of the bad loan segment, but also by the banks' capacity to grant bad loans. This capacity depends on the price of credit insurance, which in turn depends on the relative number of medium and bad entrepreneurs,  $q_M$  and  $q_B$ , and the NPV of loans. Of course, if the NPV of bad loans is only slightly negative, then banks' capacity to grant bad loans will be large. For  $n \to \infty$ , the volume of bad loans converges towards

$$\lim_{n \to \infty} L_B = \min\{q_B; \bar{L}_B\}, \quad \text{where} \quad \bar{L}_B = q_M \frac{p_M Y - r}{r - p_B Y}$$
(15)

is the banks' lending capacity to bad entrepreneurs.<sup>20</sup> Hence, the market for bad loans will *never* be saturated if the banks' lending capacity for bad loans is smaller than  $q_B$ , which happens if and only if

$$q_M (p_M Y - r) + q_B (p_B Y - r) < 0.$$
(16)

This condition implies that the aggregate NPV of all available medium and bad projects is negative; the (negative) NPV of all bad projects exceeds the (positive) NPV of all medium projects in absolute value. In reality, (16) is likely to hold. For a potential entrepreneur, it is relatively simple to come up with a negative-NPV investment, whereas it is much more difficult to find a positive-NPV project. Consequently, the potential amount of bad projects  $q_B$  will be large relative to  $q_M$ . Because the NPV of bad projects,  $p_B Y - r$ , is negative, (16) will hold. The bad market will never be saturated. This result will be important for the welfare assessment of credit risk transfer under private information.

<sup>&</sup>lt;sup>20</sup>See the Appendix for a proof of equation (15).

Welfare. When CRT is based on private information, the effects of an introduction of CRT on welfare are less clear-cut than before. The improved access to credit for medium borrowers raises welfare, but the extension of loans to negative NPV borrowers decreases welfare. Still, the overall effect on welfare is positive. Bad borrowers benefit from gaining access to credit. Medium borrowers benefit due to lower loan rates and to an improved access to loans. The banks ultimately bear the costs of the negative NPV projects of bad borrowers, but they still benefit from CRT due to the possibility of expanding in the medium market. Summing up, we have the following result.<sup>21</sup>

**Proposition 3c (Welfare)** If the CRT market does not break down, the introduction of credit risk transfer with private information strictly increases aggregate welfare. For large n, welfare converges to

$$\lim_{n \to \infty} W = q_G (Y - r) + \max\{0; q_M (p_M Y - r) + q_B (p_B Y - r)\}.$$
 (17)

Remember that, in the absence of CRT, welfare converges to  $q_G(Y - r)$  for large n. The same is true here if condition (16) holds. Hence, this proposition implies that the potential welfare gains from the introduction of CRT are completely offset at high levels of competition.

The proposition is illustrated by the third panel of Figure 6. As long as the CRT market breaks down, it does not affect welfare. The gray and the dashed welfare curves are identical for n < 14. At  $n \approx 14$ , the CRT market becomes active. Welfare jumps up, and further competition raises welfare even more. At  $n \approx 19$ , the medium market is saturated, and further competition only increases the volume of bad loans. From this point onwards, welfare decreases in competition. In the limit, the two curves (welfare without CRT and welfare with private-information CRT) converge. Again, the same results hold true also for decreasing t.

These welfare effects are interesting because they differ from the results found in the literature. In the paper by Parlour and Plantin (2008) (PP), the introduction of CRT may decrease welfare. The diverging results stem from differences in model structures. In PP, CRT affects banks in two ways. First, banks face unknown future investment opportunities (modelled through a stochastic discount factor). CRT allows banks to

<sup>&</sup>lt;sup>21</sup>As in Proposition 2c, the fact that welfare increases does not hinge upon the risk neutrality of insurers.

exploit profitable investment opportunities, which may lead to a positive welfare effect. This is comparable to the positive welfare effect of our static model, where CRT helps banks to grant additional loans that could not be financed otherwise. Second, banks have to monitor loans because firms face a moral hazard problem. If banks do not monitor, firms shirk and their projects' NPV decreases. If a bank's stakes in a firm's returns are too small (for example because the bank makes use of CRT to insure part of a loan), monitoring incentives of the bank are reduced. In order to preserve monitoring incentives, the bank must increase its relative stake in the firm. The borrowing capacity and size of the firm decline, which leads to a negative welfare effect. Hence, banks exert a negative externality on firms. Either of these two channels can dominate, which explains the ambiguous welfare result in PP. In their paper, if the welfare effect of CRT is negative, banks would like to commit to *not* using CRT instruments.

Instead of monitoring, banks screen firms in our paper before granting loans. After screening, they hold information on the firms' types that insurers do not have. As in PP, CRT distorts incentives because it induces banks to originate bad loans. However, in PP, banks harm firms by not monitoring (if this is anticipated by other investors). In our paper, the origination of bad loans is beneficial for firms (who obtain better access to loans and lower loan rates), irrelevant for insurers (who are always at their participation constraint), and beneficial for banks (who otherwise would not use CRT). Consequently, although monitoring and screening are comparable activities, the welfare effects of CRT can be quite different.

**Market Entry.** Let us now discuss the long-run effects of CRT with private information. Since the introduction of CRT increases expected profits in the short run (see Proposition 3a), it attracts more banks in the long run. Because CRT with private information increases expected profits less then CRT with public information, fewer additional banks will enter the market.

If competition is not too fierce, long-run effects of CRT reinforce short-run effects, as was the case with public information. Increasing competition through market entry further improves the access to credit for both medium and bad borrowers. However, if competition is so strong that the medium market is covered completely, market entry improves the access to credit only for bad borrowers. In this case, only the detrimental effects of CRT are reinforced. Under CRT with private information, entry is again excessive. First, as in any Salop model, firms do not take into account the negative externality that their entry has on the other firms' profits. Second, bank entry exacerbates the problem that banks lend to bad entrepreneurs to finance negative NPV projects.

## 5 Extensions

We will now briefly consider two extensions of our model. First, we analyze a modification of the Salop model where the entrepreneurs' locations can be observed by banks. Second, we will discuss what happens when insurers can observe a bank's retention of credit risk.

**Observable Locations** The Salop model assumes that banks cannot observe the entrepreneurs' locations. We will now show that this assumption is not crucial for our results. Consider the setup of Section 2, the only difference being that banks can observe the entrepreneurs' locations x. Consequently, they can (and will) price discriminate among entrepreneurs. We first examine the market for good entrepreneurs. Assume that the number of banks is so large that at least two banks compete for each entrepreneur. There is Bertrand competition for each entrepreneur. Consequently, an entrepreneur always takes the loan from the closest bank, at a loan rate that cannot be matched by the second closest bank. Take an entrepreneur at position  $x < \frac{1}{2n}$ , with the closest bank at position 0. The second closest bank is at position  $\frac{1}{n}$ , with a break-even loan rate of  $R_G = r$ . With this loan rate, the expected profit of the entrepreneur would be  $(Y - r) - t(\frac{1}{2n} - x)$ . Hence, the closest bank must offer a rate such that

$$(Y - R_G) - t x = (Y - r) - t \left(\frac{1}{2n} - x\right),$$
  

$$R_G = r + t \left(\frac{1}{n} - 2x\right).$$
(18)

Each bank can extract exactly the difference in transportation costs from an entrepreneur. The bank's clientele reaches from  $-\frac{1}{2n}$  to  $\frac{1}{2n}$ , hence aggregate profits from *G*-type entrepreneurs are

$$\Pi_G = 2 \int_0^{\frac{1}{2n}} \left[ r + t \left( \frac{1}{n} - 2x \right) \right] dx = \frac{t}{2n^2}.$$
(19)

In analogy to (1),  $\Pi_G - r l_M \ge 0$  must hold. In equilibrium,  $l_M = \Pi_G/r$ . Consequently, we immediately get an analogy to Proposition 1b: As the number of banks n increases, profits in the good loan segment decrease; hence banks have less money to put at risk in the medium loan segment, hence the market penetration in the medium segment decreases. This mechanism is the driving force of the main results of this paper. Therefore, the results of the paper will also go through with observable locations. The comparative statics with respect to the degree of competition are identical to those with unobservable locations of entrepreneurs. Especially welfare can again be non-monotonic in the case of private-information CRT.

**Observable Retention Rates** In the private information case, we have based our calculations on the assumption that insurers can observe neither the type of an entrepreneur, nor the fraction of a loan's risk that is retained in a bank's portfolio. As a consequence, the retention rate was zero for bad loans, and  $1 - \kappa$  for medium loans (and 1 for good loans). With an observable retention rate, banks can commit to granting and insuring only medium loans. In equilibrium, the retention rate has to be high enough to destroy a bank's incentives to grant bad loans and then insure them.

The expected profit from a bad loan, insured at the premium of a medium loan, is  $\kappa p_M R_M + (1 - \kappa) p_B R_M - r$ . The profit from granting and insuring bad loans has to be negative,

$$\kappa p_M R_M + (1-\kappa) p_B R_M - r < 0,$$

defining a minimum retention rate  $1 - \kappa$ , where

$$\kappa = \frac{r/R_M - p_B}{p_M - p_B}.$$

Furthermore, a modified version of (1) holds. A bank's profits from good loans are unaffected and are equal to  $\Pi_G = q_G t/n^2$ . Now there is a multiplier effect. Banks can use the profits from good loans to grant medium loans, of which they insure a fraction  $\kappa$ . The insured fraction leads to new safe cash streams, which again can be used to grant medium loans. Since the multiplier is smaller than one, the equilibrium volume of medium loans can fall short of a bank's desired volume. The shadow price  $\lambda$  will then still be positive.

Note that observable retention rates can prevent the granting of bad loans and will therefore increase welfare relative to the situation considered before. However, rising competition will still be welfare-decreasing for large n because fiercer competition reduces banks' risk-bearing capacities (but less than without CRT). Therefore, welfare will be lower than under public information, especially for high levels of competition. In the limiting case of extreme competition  $(n \to \infty)$ , banks' profits in the good loan segment vanish. The volume of medium loans converges to zero, and welfare drops to  $W = q_G (Y - r)$ , as in Proposition 3c.

# 6 Conclusion

This paper has shown how credit risk transfer can improve the access to finance for risky borrowers by increasing banks' risk-bearing capacities and thereby relaxing lending constraints. Without CRT, a bank may be reluctant to grant loans to risky borrowers because such loans threaten its solvency. An introduction of markets for CRT generally leads to a loan expansion and thereby to an increase in welfare because it enables borrowers to finance profitable projects that would otherwise not have been carried out.

However, a bank's ability to transfer risks depends on whether the bank grants loans on the basis of public or private information. If loans are granted on the basis of public information, credit risk transfer works smoothly because banks can easily convey the quality of their borrowers to insurers and will not have an incentive to grant (and transfer) unprofitable loans. If loans are granted on the basis of private information, the transfer of credit risk is more difficult because the insurers cannot observe the quality of a bank's borrowers. This leads to a moral hazard problem at the originating bank. It can exploit the informational asymmetry by granting unprofitable loans and transferring the risk to the insurers. The insurers anticipate this and demand a lemons premium. As a consequence, banks do not insure their loan portfolio to the same degree as with public information. Here the possibility of transferring risks improves the access to finance not only for medium (i. e. risky, but profitable) borrowers, but also for bad (i. e. unprofitable) borrowers, which leads to an increase in aggregate risk in the economy.

Nevertheless, the overall welfare effect of an introduction of CRT markets is always positive in our setup. Even with private information, CRT is beneficial because the welfare gains from the improved access to credit for medium borrowers overcompensate the welfare losses from the improved access to credit for bad borrowers. Furthermore, we have emphasized the role of banking competition. In our basic setup, we find that the undersupply of risky loans is most severe in a highly competitive environment. The reason is that the banks' margins in other types of business (the good loan segment) and hence the potential to absorb losses from risky loans are small under such circumstances. With CRT, increasing competition no longer leads to a deterioration in borrowers' access to finance; in fact, the borrowers' access to finance improves both with public and private information. However, increasing competition may be harmful in the presence of CRT with private information. When the medium loan market is saturated, a further increase in competition improves the access to finance only for bad borrowers, and welfare decreases. In the limit, this completely offsets the welfare gains from credit risk transfer.

CRT markets break down when banking competition is very low. Low competition implies that banks have large risk-bearing capacities and can keep most of their medium loans in their own balance sheets. This reduces the average quality of loans to be insured, such that insurers are no longer willing to participate. This may explain why CRT markets appeared in an environment of intensifying banking competition, after banking markets had been deregulated in the 1980s and 1990s and banks' margins had been reduced by competition from non-bank intermediaries.

The sharp distinction between banking markets with public and private information should not be taken too literally. The pure public information case can hardly be found in the real world and should rather be seen as a benchmark case in which CRT markets work perfectly. In reality, loan markets are always characterized by some degree of asymmetric information; hence, CRT markets will never work perfectly and will always involve some moral hazard in the origination of loans. Moreover, the welfare analysis presented in this paper abstracts from the costs arising from financial instability. Such costs would have to be balanced against the benefits from CRT, such as the improved access to finance for profitable borrowers.

Our results on the destabilizing role of highly competitive banking markets in the presence of credit risk transfer are well in line with the traditional literature on the harmful effects of banking competition on financial stability. Our paper yields a number of interesting testable implications regarding the effect of competition on banking markets. First, CRT markets are most likely to develop in an environment of intensive banking competition. Second, in the presence of CRT with private information, a rise in competition tends to lower average loan quality when most profitable lending opportunities have already been exploited; then new loans tend to be low quality loans. Both predictions are in line with the observations from the current subprime crisis. The development of CRT markets coincided with intensifying competition in the banking market due to deregulation and the entry of non-bank competitors. Moreover, the late years of the credit boom preceding the crisis were characterized by both increasing competition in the banking sector and decreasing loan quality. Further research is needed to explore in more detail the relationship between banking competition and financial stability in the presence of CRT markets.

## A Appendix

### A.1 Proofs

**Proof of Proposition 1a.** We will first solve for the endogenous variables of our model in an equilibrium where the good market is covered, the medium market is not covered, and the shadow price  $\lambda$  is strictly positive. Then we will derive the conditions under which this equilibrium obtains.

Banks maximize their expected profits given by (6), subject to condition (1). We set up the Lagrangian  $\Lambda$  and plug in the loan demand functions  $l_G$  and  $l_M$  from (3) and (5),

$$\Lambda = (R_G - r) l_G + (p_M R_M - r) l_M + \lambda \left[ (R_G - r) l_G - r l_M \right] = (R_G - r) q_G \left( \frac{1}{n} + \frac{R'_G - R_G}{t} \right) + (p_M R_M - r) 2 p_M q_M \frac{Y - R_M}{t} + \lambda \left[ (R_G - r) q_G \left( \frac{1}{n} + \frac{R'_G - R_G}{t} \right) - r 2 p_M q_M \frac{Y - R_M}{t} \right].$$
(20)

Taking the first-order conditions of the Lagrangian  $\Lambda$  with respect to  $R_G$ ,  $R_M$ , and  $\lambda$ , and setting  $R'_G = R_G$  in a symmetric equilibrium, we obtain

$$R_G = r + \frac{t}{n},\tag{21}$$

$$R_M = Y - \frac{q_G t^2}{2 n^2 p_M q_M r}, \text{ and}$$
(22)

$$\lambda = \frac{p_M Y}{r} - 1 - \frac{q_G t^2}{n^2 q_M r^2},\tag{23}$$

where  $\lambda$  is the shadow price of constraint (1). Equilibrium loan volumes and profits are

$$l_G = \frac{q_G}{n}, \quad L_G = n \, l_G = q_G, \tag{24}$$

$$l_M = \frac{q_G t}{n^2 r}, \quad L_M = n \, l_M = \frac{q_G t}{n \, r}, \text{ and}$$
 (25)

$$\Pi = \frac{q_G t}{2 n^4 q_M r^2} \left( 2 n^2 p_M q_M r Y - q_G t^2 \right).$$
(26)

We now show under which conditions this equilibrium obtains. We first derive the condition under which the good market is covered for lower n than the medium market. Consider a situation in which the market for good loans is not covered. Then banks enjoy local monopolies in this loan segment. The critical, indifferent entrepreneur at position  $x_G$  would have  $(Y - R_G) - t x_G = 0$ , hence  $x_G = (Y - R_G)/t$  and  $l_G = 2 q_G (Y - R_G)/t$ . Denote by  $\Pi_G$  a bank's profits in the good loan segment. Then  $\Pi_G = (R_G - r) l_G = (R_G - r)(Y - R_G) 2 q_G/t$ . The first-order condition  $\partial \Pi_G / \partial R_G = 0$ implies that  $R_G = (Y+r)/2$ ,  $l_G = q_G (Y-r)/t$ , and  $\Pi_G = q_G (Y-r)^2/(2t)$ . Since there is no competition among banks, these values do not depend on n. Equation (1) is equivalent to  $\Pi_G = r l_M$ , which implies that  $l_M = \Pi_G/r$  if (1) binds. Hence, as long as banks do not interact at all,  $L_G = n l_G$  and  $L_M = n l_M$  are proportional to n.

The question is now under which condition the good market is covered  $(L_G = q_G)$  for lower *n* than the medium market. Clearly, the answer must depend on the ratio between  $q_G$  and  $q_M$ . If the good market is relatively small, the banks' risk-bearing capacity will be small and banks can cover only part of the medium market. Then the good market will be saturated first. The good sector is saturated if  $L_G = n q_G (Y - r)/t = q_G$ , hence if n = t/(Y - r). The medium sector is saturated if  $L_M = n q_G (Y - r)^2/(2tr) = q_M$ , hence if  $n = 2 q_M r t/[q_G (Y - r)^2]$ . Comparing these two critical values for *n*, one finds that the good segment is saturated for a lower *n* than the medium segment if

$$\frac{q_M}{q_G} > \frac{Y - r}{2r}.\tag{27}$$

This yields the first condition of Proposition 1a. Note that this condition is very likely to be satisfied in reality. The fraction (Y - r)/r gives the return of a project in the good segment. This number will not exceed a few percentage points; the right hand side of (27) will thus typically be well below 1. Hence,  $q_M$  could even be well below  $q_G$  under this condition.

For  $n \ge t/(Y - r)$ , the good market is covered. This is the second condition in the proposition. If this condition were violated, banks would not compete at all. All interest

rates would be independent of n, and aggregate loan volumes (and welfare) would be proportional to n. At the critical n = t/(Y - r), the medium market is not covered due to condition (27). For larger n,  $L_M$  decreases (see (25)). Hence, the medium market is not covered.

Finally, condition (1) binds in equilibrium if  $\lambda > 0$ , hence if  $n > t \sqrt{q_G} / \sqrt{q_M r (p_M Y - r)}$ . This is the third condition in the proposition.

**Proof of Proposition 1b.** We see that, for an individual bank,  $dl_G/dn < 0$  and  $dl_M/dn < 0$ , see (24) and (25). In the aggregate,  $L_G = q_G$ . The market is still covered completely, such that  $dL_G/dn = 0$ , which proves the first statement of the proposition. In the medium market,  $L_M = (q_G t)/(n r)$ , see (25), which depends negatively on n. This proves the second statement of the proposition. Finally, taking the derivative of (23) with respect to n, we find that  $d\lambda/dn > 0$ , proving the third statement of the proposition.  $\Box$ 

**Proof of Proposition 1c.** Welfare consists of the expected profits and rents of banks, depositors, and borrowers. Depositors' participation constraints are binding, their rents are equal to zero. Aggregate welfare for good and medium entrepreneurs is

$$W_G = q_G 2n \int_0^{\frac{1}{2n}} ((Y - R_G) - tx) \, dx = q_G \left(Y - r - \frac{5t}{4n}\right), \text{ and}$$
(28)

$$W_M = q_M 2n \int_0^{x_M} (p_M (Y - R_M) - tx) \, dx = \frac{q_G^2 t^3}{4 n^3 q_M r^2},\tag{29}$$

where  $x_M$  is defined by (4). The banks' profits  $\Pi$  are given by (6). Aggregate welfare is then

$$W = W_G + W_M + n \Pi = q_G \left( (Y - r) + t \frac{p_M Y - 5/4r}{n r} - \frac{q_G t^3}{4 q_M n^3 r^2} \right).$$
(30)

The first-order condition with respect to n yields

$$\frac{\partial W}{\partial n} = q_G \left( -t \, \frac{p_M \, Y - 5/4 \, r}{n^2 \, r} + 3 \, \frac{q_G \, t^3}{4 \, q_M \, n^4 \, r^2} \right) = 0$$

Solving for n gives the optimal  $n^*$  as in (7). For  $n \to \infty$ , W converges to  $q_G(Y-r)$ .

**Proof of Proposition 2a.** The bank's profit function is given by (9). Banks can swap risky loans against safe payments with the same expected return by insuring a loan, hence condition (1) is no longer binding. This implies that  $\lambda$  is equal to 0, proving the third statement of the proposition.

Loan volumes are as in (3) and (5), given that the medium loan market is not covered completely. Plugging in loan volumes, taking the derivative of a bank's expected profits  $\Pi$ with respect to  $R_G$  and  $R_M$ , and taking into account the symmetry of banks ( $R'_G = R_G$ ), we obtain

$$R_G = r + \frac{t}{n}$$
 and  $R_M = \frac{Y}{2} + \frac{r}{2p_M}$ . (31)

Hence,  $R_G$  is unchanged by the introduction of CRT, and the market is covered completely as before,  $L_G = q_G$ . Hence,  $dL_G/dn = 0$ , which proves the first statement of the proposition.

Comparing  $R_M$  before and after the introduction of CRT, we find that  $R_M$  decreases due to the introduction of CRT markets if and only

$$\begin{aligned} \frac{Y}{2} + \frac{r}{2 p_M} &< Y - \frac{q_G t^2}{2 n^2 p_M q_M r}, \\ \frac{r}{p_M} &< Y - \frac{q_G t^2}{n^2 p_M q_M r}, \\ 0 &< \frac{p_M Y}{r} - 1 - \frac{q_G t^2}{n^2 q_M r^2}, \end{aligned}$$

hence if  $\lambda$  is positive in the absence of CRT, see (23). Therefore, individual and aggregate loan volumes in the medium loan segment,  $l_M$  and  $L_M$  increase, see (5), which proves the second statement of the proposition.

Finally, expected profits are

$$\Pi = q_G \frac{t}{n^2} + q_M \frac{(p_M Y - r)^2}{2t}.$$
(32)

The condition  $\lambda > 0$  is sufficient for an increase in expected profits due to the introduction of CRT. This proves the final statement of the proposition.

The expansion of medium loans due to the introduction of CRT may lead to a complete coverage of the medium loan segment. As before, the market for good loans is not affected by CRT. Since the medium market is covered, banks no longer enjoy local monopolies, but they compete à la Salop, as in the good loan segment. Banks maximize  $\Pi_M = (p_M R_M - r) l_M$  subject to  $l_M = q_M (1/n + (R'_M - R_M)/t)$ , in analogy to (3). Taking the first-order condition, setting  $R'_M = R_M$  due to symmetry, and solving for  $R_M$ , we obtain

$$R_M = \frac{r}{p_M} + \frac{t}{n}.$$

Loan volumes are  $l_M = q_M/n$  due to symmetry, banks' profits are  $\Pi_M = p_M q_M t/n^2$ .

Let us again go through the statements of the proposition. First, the amount of good loans is unaffected. Second, the amount of medium loans has increased (the market was not covered before). Third, condition (1) is not binding, hence the shadow price  $\lambda$  is zero. Finally, profits increase if condition (1) was binding in the absence of CRT. Hence, all four statements of the proposition are also true when the medium market is covered after the introduction of CRT.

**Proof of Proposition 2b.** As argued above, the aggregate loan volume in the good loan segment,  $L_G = q_G$ , does not depend on n, as the market is already saturated. This gives us the first part of the proposition. Moreover, we can plug  $R_M$  from (31) in (5) to obtain  $l_M = q_M (p_M Y - r)/t$ . We see that the medium loan volume of a single bank does not depend on the number of banks. As a result,  $L_M = n l_M$  increases monotonically in n until the medium loan market is covered completely. This proves the second part of the proposition.

**Proof of Proposition 2c.** Welfare now consists of the expected profits and rents of banks, depositors, borrowers, and insurers. Risk insurance is fair, hence insurers' expected profits are zero. Depositors' rents are also zero. Consequently,  $W = W_G + W_M + n \Pi$ , as before. If the market for medium loans is not covered completely, we obtain the following results.  $W_G$  is as in (28),

$$W_{M} = q_{M} 2 n \int_{0}^{x_{M}} (p_{M} (Y - R_{M}) - t x) dx = n q_{M} \frac{(p_{M} Y - r)^{2}}{4 t}, \text{ and}$$
$$\Pi = q_{G} \frac{t}{n^{2}} + q_{M} \frac{(p_{M} Y - r)^{2}}{2 t}, \text{ hence}$$
$$W = q_{G} \left( (Y - r) - \frac{t}{4 n} \right) + q_{M} \frac{3 n (p_{M} Y - r)^{2}}{4 t}.$$
(33)

For  $n = t/(p_M Y - r)$ , the medium market is covered. The welfare components are then

$$W_M = q_M \left( (p_M Y - r) - \frac{5t}{4n} \right) \text{ and}$$
  

$$\Pi = (q_G + q_M) \frac{t}{n^2}, \text{ hence}$$
  

$$W = q_G \left( (Y - r) - \frac{t}{4n} \right) + q_M \left( (p_M Y - r) - \frac{t}{4n} \right). \tag{34}$$

The aggregate deposit supply is perfectly elastic, hence depositors' expected utility does not change although the aggregate deposit volume increases due to the introduction of CRT. For good borrowers, loan rates  $R_G$  remain unchanged. CRT does not help bad borrowers to gain access to credit. Medium borrowers, however, profit from the introduction of CRT in two ways. Those borrower who already had access to credit in the absence of CRT benefit from lower loan rates  $R_M$ . Additionally, the volume  $L_M$  expands; some medium borrowers gain access to credit due to CRT. Finally, consider the banks' expected profits. Introducing CRT, banks' profits in the good loan segment do not change; expected profits from the medium loan segment increase. Consequently, aggregate welfare increases.

This can also be shown formally. Comparing (30) with (33), we obtain

$$\begin{split} q_{G}\Big((Y-r) - \frac{t}{4n}\Big) + q_{M}\frac{3n(p_{M}Y-r)^{2}}{4t} > q_{G}\Big((Y-r) + t\frac{p_{M}Y-5/4r}{nr} - \frac{q_{G}t^{3}}{4q_{M}n^{3}r^{2}}\Big) \\ - \frac{t}{4n} + \frac{q_{M}}{q_{G}}\frac{3n(p_{M}Y-r)^{2}}{4t} > t\frac{p_{M}Y-5/4r}{nr} - \frac{q_{G}t^{3}}{4q_{M}n^{3}r^{2}} \\ \frac{q_{M}}{q_{G}}3n^{4}(p_{M}Y-r)^{2} - n^{2}t^{2} > t^{2}\frac{n^{2}(4p_{M}Y-5r)}{r} - \frac{q_{G}t^{4}}{q_{M}r^{2}}. \end{split}$$

Solving for n, we get

$$n > t \frac{\sqrt{q_G}}{\sqrt{q_M r \left(p_M Y - r\right)}},$$

which is the condition for a positive shadow price  $\lambda$  at the end of Proposition 1a. Analogously, a comparison of (30) and (34) shows that, in the case of saturated medium loan markets, welfare increases through the introduction of CRT markets for any positive n.

Finally, we see immediately that both welfare functions, (33) and (34), are strictly increasing in n. However, welfare increases more slowly once the medium market is covered. For large n, the medium market is eventually saturated, and welfare in (34) converges to  $q_G(Y-r) + q_M(p_M Y - r)$ .

**Proof of Proposition 3a.** We already noted that profit-maximizing banks that respect the constraint (11) set identical loan rates for medium and bad entrepreneurs, as given by (14). For good entrepreneurs, the loan rate is  $R_G = r + t/n$ , as in (21). Furthermore, in (14),  $\bar{p}$  is determined by  $\beta$ , and  $\beta$  is determined by  $\kappa$  in (13). Hence, the symmetric perfect Bayesian equilibrium is defined by (21) and (14), reflecting the profit maximizing behavior of banks, and (13), reflecting the rational beliefs of the insurers. Consequently, we have three equations for three variables,  $R_G$ ,  $R = R_M = R_B$ , and  $\bar{p}$  (the last of which defines  $\kappa$ ).

We can solve this system of equations for  $R_G$ , R, and  $\bar{p}$ . We find two solutions for  $\bar{p}$ ,

$$\bar{p} = \frac{n^2 \left(p_B \, q_B + p_M \, q_M\right) r \left(p_M \, Y + r\right) - p_M \, q_G \, t^2 \pm \sqrt{A}}{n^2 \, Y[\, p_B \, q_B \, Y \, (p_M - p_B) + 2 \, r \, (p_M \, q_M + p_B \, q_B)] - 2 \, q_G \, t^2}, \text{ where}$$
(35)  
$$A = p_M^2 \, q_G^2 \, t^4 + n^4 \, r^2[\, p_M \, q_M \, (p_M \, Y - r) + p_B \, q_B (p_B \, Y - r)]^2 - 2 \, n^2 \, q_G \, r \, t^2[\, p_B \, q_B \, (p_M^2 \, Y - p_B r) + p_M \, q_M \, (p_M^2 \, Y - p_M \, r)].$$

Hence, there are also two solutions for R, whereas  $R_G = r + t/n$  in both cases. There are two equilibria with different values for R and  $\bar{p}$ , but identical  $R_G$ . The existence of multiple equilibria is intuitive. Assume that  $\bar{p}$ , the success probability expected by insurers, is relatively low. Then the insurers will demand a high premium for insuring credit risk. Therefore, banks have high marginal costs of granting medium and bad loans, and banks will pass on part of these costs to the entrepreneurs.  $R = R_M = R_B$  will be high, and the loan volumes  $l_M$  and  $l_B$  will be low. However, the profits from good loans do not depend on  $\bar{p}$ . Consequently, a bank can keep a larger fraction of medium loans on its books, so that  $\kappa$  is relatively low, and the average quality of insured loans is also low (implying a low  $\bar{p}$ ). Hence, the insurers' initial belief is justified in equilibrium. An analogous argument can be made for relatively large  $\bar{p}$ . Hence, the existence of multiple equilibria is intuitive, and it is confirmed by the algebraic expression of (35).

However, the equilibrium with the larger  $\bar{p}$  is Pareto-efficient. Insurers are indifferent between the two equilibria. Entrepreneurs prefer the equilibrium with higher loan volumes and lower loan rates, i.e. that with higher  $\bar{p}$ . Banks prefer the equilibrium with lower credit insurance premia, i.e. that with higher  $\bar{p}$ . Consequently, in the following, we will concentrate on the Pareto-efficient equilibrium, that with the positive sign in (35).

Some statements of Proposition 3a follow immediately. *First* statement:  $R_G$  is unaffected by CRT; the determining equation is the same as in the absence of CRT. Consequently,

the coverage of the good loan segment does not change,  $L_G = q_G$ . Third statement: The amount of bad loans is strictly positive under private-information CRT, hence it increases due to CRT. Fifth statement: Expected profits of banks increase less than under public-information CRT because banks cannot commit to not granting loans to bad borrowers. In equilibrium, insurers will demand higher premia for insuring loans than with public-information CRT. Consequently, the costs of providing an additional loan are higher. Banks ultimately bear the costs arising from the lemons problem. Second statement: Banks' eagerness to lend to medium entrepreneurs is determined by marginal costs. These consist of refinancing costs r and of the costs from CRT (banks always insure the risk of the marginal medium loan). The latter costs are determined by  $\beta$ . The higher the probability  $\beta$  that an insured loan is medium, the lower are the costs of insurance. With public-information CRT,  $\beta = 1$  because no bad loans are granted; with private-information CRT,  $\beta < 1$ . As a consequence, the marginal costs are higher with private-information CRT, the loan volume  $l_M$  is smaller, and the loan rate  $R_M$  is higher. However, with a volume  $l_M$  like in the absence of CRT, banks do not need CRT. Hence with a positive volume of CRT,  $l_M$  must have increased. Fourth statement:  $\lambda$  is the shadow price of condition (11), which is identical to (1) for  $l_B = 0$ . A marginal increase in the capital buffer increases expected profits by  $\lambda$ . This increase in expected profits is due to the bank's ability to expand lending to medium (and bad) borrowers, and it hence depends on  $R_M$  (and  $R_B$ ). Since  $R_M$  is larger under private-information CRT than under public-information CRT, the shadow price  $\lambda$  must be larger. In comparison to the case without CRT, however,  $\lambda$  must decrease. As the volume of loans  $l_M$  and  $l_B$  increases, the according interest rate R falls, hence the shadow price of the binding condition (11) is smaller than in the absence of CRT. 

**Proof of Proposition 3b.** Consider equation (11),

$$(R_G - r) l_G + \bar{p} R (l_B + \kappa l_M) - (l_M + l_B) r \ge 0,$$

and let the number of banks n increase. Competition and loan rates on the market for good loans are the same as in the absence of CRT (and as with public information). Hence, the good loan market is completely covered as before,  $dL_G/dn = 0$ , which proves the first statement of the proposition. An increase in n implies that  $(R_G - r) l_G$  decreases;  $R_G$  decreases,  $l_G$  decreases, and r remains constant. Let us keep R fixed for the moment. Then, both  $l_M$  and  $l_B$  do not change as n increases. Hence, banks need to raise  $\kappa$  in order to fulfill (11). (13) reveals that  $\beta$  increases when a larger fraction of medium loans is insured (with constant loan volumes). Insurers anticipate a better quality of insured loans, and  $\bar{p} = \beta p_M + (1 - \beta) p_B$  increases as well. This means that the marginal profit from an insured loan,  $\bar{p}R - r$ , increases. Consequently, banks will expand loan volumes  $l_M$  and  $l_B$  by lowering loan rates  $R_M = R_B = R$ . This triggers off a reinforcing multiplier effect. Due to (11), banks need to increase  $\kappa$  even further. The final equilibrium has higher aggregate loan volumes  $L_M = n l_M$  and  $L_B = n l_B$  at lower loan rates R, and a higher share of insured medium loans  $\kappa$ . Of course,  $L_M$  and  $L_B$  can only rise until the loan segments are covered completely.

All these comparative statics hold only if the market for CRT does not break down. Mathematically, this happens if the term under the square root in (35), A, becomes negative, such that no real solution obtains. The critical  $\bar{n}$  below which the market for CRT breaks down is hence defined implicitly by

$$\bar{n}^4 r^2 \left[ p_B q_B \left( p_B Y - r \right) + p_M q_M \left( p_M Y - r \right) \right]^2 -2 \bar{n}^2 q_G r t^2 \left[ p_B q_B \left( p_M^2 Y - p_B r \right) + p_M^2 q_M \left( p_M Y - r \right) \right] = p_M^2 q_G^2 t^4.$$
(36)

Below this  $\bar{n}$ , all parameters are as in Section 3 (no CRT).

**Proof of Equation (15).** When the number of banks n increases, the bad loan segment will not necessarily be covered completely. To show this, we will derive the equilibrium where the bad loan segment is never covered and will then derive the condition, under which this equilibrium obtains.

If n goes to  $\infty$ , the profits from good loans go to zero. Consequently, banks must insure all loans that they grant,  $\kappa = 1$ . In an equilibrium where the bad market is not covered even for large n, banks always enjoy local monopolies over bad entrepreneurs. The expected profit on bad loans is

$$l_B (\bar{p} R_B - r) = 2 q_B p_B \frac{\bar{p} R_B - r}{\bar{p} t} (Y - R_B),$$

which is maximized for

$$R_B = \frac{\bar{p}\,Y - r}{2\,\bar{p}},$$

which yields an aggregate bad loan volume of

$$L_B = n q_B p_B \frac{\bar{p} Y - r}{\bar{p} t}.$$

The aggregate medium loan volume is  $L_M = q_M$ . Considering that  $\beta = \kappa l_M / (\kappa l_M + l_B)$ ,  $\kappa = 1$ , and  $l_M = q_M / n$ , we obtain

$$\bar{p} = \beta \, p_M + (1 - \beta) \, p_B = \frac{\bar{p} \, p_M \, q_M \, t + n \, p_B^2 \, q_B \, (\bar{p} \, Y - r)}{\bar{p} \, q_M \, t + n \, p_B \, q_B \, (\bar{p} \, Y - r)}.$$

The solution for  $\bar{p}$  is

$$\bar{p} = \frac{Z \pm \sqrt{Z^2 - 4 n r p_B^2 q_B X}}{2 X},$$

where  $X = q_M t + n q_B p_B Y$  and  $Z = n p_B q_B (p_B Y + r) + p_M q_M t$ . For large  $n \to \infty$ , we obtain  $X \approx n q_B p_B Y$  and  $Z \approx n p_B q_B (p_B Y + r)$ . *n* drops out of the expression for  $\bar{p}$ , hence now

$$\bar{p} = \frac{p_B q_B (p_B Y + r) \pm [p_B q_B (P_B Y - r)]}{2 p_B q_B Y}.$$

The two solutions are  $\bar{p} = p_B$  and  $\bar{p} = Y/r$ , the first of which is economically meaningless (it would imply that only bad loans are insured, but in this case, CRT would be prohibitively expensive).

With  $\kappa = 1$ , we have  $\beta = l_M/(l_M + l_B)$  and  $L_B = n \cdot l_B = \frac{n(1-\beta)}{\beta} l_M$ . Considering that  $\bar{p} = r/Y = \beta p_M + (1-\beta) p_B$  and  $l_M = q_M/n$ , we obtain

$$L_B = q_M \frac{p_M Y - r}{r - p_B Y}.$$
(37)

This volume can only be reached if it does not exceed  $q_B$ . This will happen if condition (16) holds; then, the market for bad loans is never covered completely, and the maximum volume of bad loans is as calculated above. If condition (16) does *not* hold, all bad entrepreneurs will get loans if competition is strong enough.

**Proof of Proposition 3c.** Aggregate welfare consists of the expected profits and utilities of banks, depositors, (good, medium, and bad) borrowers, and insurers. Depositors, good borrowers, and insurers are unaffected by the introduction of private-information CRT; banks and medium borrowers profit. Given that CRT is now based on private information, loans are also granted to bad borrowers with negative NPV projects. Bad borrowers profit from the improved access to credit. Summing up, the introduction of CRT increases aggregate welfare in spite of the expansion of negative NPV loans.

Now consider the limit of  $n \to \infty$ . The consequences of equation (15) for welfare are immediate. For  $n \to \infty$ , aggregate transportation costs vanish, and aggregate welfare

equals the aggregate NPV of projects. If condition (16) holds, aggregate welfare is thus

$$W = L_G(Y - r) + L_M(p_M Y - r) + L_B(p_B Y - r)$$
  
=  $q_G(Y - r) + q_M(p_M Y - r) + q_M \frac{p_M Y - r}{r - p_B Y}(p_B Y - r)$   
=  $q_G(Y - r)$ .

The welfare gain from loans to medium entrepreneurs due to CRT is completely offset by the negative NPV of bad projects for high levels of competition.

If condition (16) does not hold, aggregate welfare is simply

$$W = q_G (Y - r) + q_M (p_M Y - r) + q_B (p_B Y - r).$$

All markets are covered for large n, hence the only effect of an increase in n is a reduction of transportation costs.

### References

- ALLEN, F., AND D. M. GALE (2005): "Systemic Risk and Regulation," in *The Risks of Financial Institutions*, ed. by M. Carey, and R. M. Stulz, pp. 341–375. The University of Chicago Press.
- ASHCRAFT, A., AND J. SANTOS (2008): "Has the CDS Market Lowered the Cost of Corporate Debt?," Working Paper, SSRN.
- CERASI, V., AND J.-C. ROCHET (2008): "Solvency Regulation and Credit Risk Transfer," Working Paper, SSRN.
- CHIESA, G. (2008): "Optimal Credit Risk Transfer, Monitored Finance, and Banks," Journal of Financial Intermediation, 17(4), 464–477.
- DEGRYSE, H., AND S. ONGENA (2005): "Distance, Lending Relationships, and Competition," *Journal of Finance*, 60(1), 231–266.
- DELL'ARICCIA, G., D. IGAN, AND L. LAEVEN (2008): "Credit Booms and Lending Standards: Evidence from the Subprime Mortgage Market," IMF Working Paper WP/08/106.

- DUFFIE, D. (2007): "Innovations in Credit Risk Transfer: Implications for Financial Stability," BIS Working Paper No. 255.
- FREIXAS, X., AND J.-C. ROCHET (1997): *Microeconomics of Banking*. Cambridge University Press, Cambridge (MA).
- GORTON, G. B., AND G. G. PENNACCHI (1995): "Banks and Loan Sales Marketing Nonmarketeable Assets," *Journal of Monetary Economics*, 35, 389–411.
- KEYS, B., T. MUKHERJEE, A. SERU, AND V. VIG (2009a): "Did Securitization Lead to Lax Screening? Evidence from Subprime Loans," *Quarterly Journal of Economics*, forthcoming.
- (2009b): "Financial Regulation and Securitization: Evidence from Subprime loans," Working Paper, SSRN.
- KLEIN, M. A. (1973): "A Theory of the Banking Firm," Journal of Money, Credit, and Banking, 3(2), 205–218.
- MONTI, M. (1972): "Deposits, Credit, and Interest Rate Determination under Alternative Bank Objectives," in *Mathematical Methods in Investment and Finance*, ed. by G. Szegő, and K. Shell, p. 431–454. North-Holland.
- MORRISON, A. D. (2005): "Credit Derivatives, Disintermediation, and Investment Decisions," *Journal of Business*, 78(2), 621–647.
- PARLOUR, C. A., AND G. PLANTIN (2008): "Loan Sales and Relationship Banking," Journal of Finance, 63(3), 1291–1314.
- PENNACCHI, G. G. (1988): "Loan Sales and the Cost of Bank Capital," *Journal of Finance*, 43(2), 375–396.
- PETERSEN, M. A. (2004): "Information: Hard and Soft," Mimeo Kellogg School of Management.
- SALOP, S. C. (1979): "Monopolistic Competition with Outside Goods," Bell Journal of Economics, 10(1), 141–156.
- SHUBIK, M., AND R. LEVITAN (1980): *Market Structure and Behaviour*. Harvard University Press, Cambridge (MA).

- STEIN, J. (2002): "Information Production and Capital Allocation: Decentralized versus Hierarchical Firms," *Journal of Finance*, 57(5), 1891–1922.
- VICKERY, J. (2007): "How Do Financial Frictions Shape the Product Market? Evidence from Mortgage Originations," Working Paper.
- WAGNER, W., AND I. W. MARSH (2006): "Credit Risk Transfer and Financial Sector Performance," *Journal of Financial Stability*, 2(2), 173–193.

### Preprints 2009

- 2009/32: Jansen J., Beyond the Need to Boast: Cost Concealment Incentives and Exit in Cournot Duopoly
- 2009/31: Fellner G., Sausgruber R., Traxler C., Testing Enforcement Strategies in the Field: Legal Threat, Moral Appeal and Social Information
- 2009/30: Lüdemann J., Rechtsetzung und Interdisziplinarität in der Verwaltungsrechtswissenschaft

forthcoming in: Öffentliches Recht und Wissenschaftstheorie, Funke A., Lüdemann J., (Eds.), Tübingen, Mohr Siebeck, pp. 125-150

- 2009/29: Engel C., Rockenbach B., We Are Not Alone: The Impact of Externalities on Public Good Provision
- 2009/28: Gizatulina A., Hellwig M., Informational Smallness and the Scope for Limiting Information Rents
- 2009/27: Hahmeier M., Prices versus Quantities in Electricity Generation
- 2009/26: Burhop C., The Transfer of Patents in Imperial Germany
- 2009/25: Burhop C., Lübbers T., The Historical Market for Technology Licenses: Chemicals, Pharmaceuticals, and Electrical Engineering in Imperial Germany
- 2009/24: Engel C., Competition as a Socially Desirable Dilemma Theory vs. Experimental Evidence
- 2009/23: Morell A., Glöckner A., Towfigh E., Sticky Rebates: Rollback Rebates Induce Non-Rational Loyalty in Consumers Experimental Evidence
- 2009/22: Traxler C., Majority Voting and the Welfare Implications of Tax Avoidance
- 2009/21: Beckenkamp M., Engel C., Glöckner A., Irlenbusch B., Hennig-Schmidt H., Kube S., Kurschilgen M., Morell A., Nicklisch A., Normann H., Towfigh E., Beware of Broken Windows! First Impressions in Public-good Experiments
- 2009/20: Nikiforakis N., Normann H., Wallace B., Asymmetric Enforcement of Cooperation in a Social Dilemma

forthcoming in: Southern Economic Review, In Press.

- 2009/19: Magen S., Rechtliche und ökonomische Rationalität im Emissionshandelsrecht
- 2009/18: Broadberry S.N., Burhop C., Real Wages and Labour Productivity in Britain and Germany, 1871-1938: A Unified Approach to the International Comparison of Living Standards
- 2009/17: Glöckner A., Hodges S.D., Parallel Constraint Satisfaction in Memory-Based Decisions
- 2009/16: Petersen N., Review Essay: How Rational is International Law?

forthcoming in: European Journal of International Law, vol. 20, In Press.

- 2009/15: Bierbrauer F., On the legitimacy of coercion for the financing of public goods
- 2009/14: Feri F., Irlenbusch B., Sutter M., Efficiency Gains from Team-Based Coordination Large-Scale Experimental Evidence
- 2009/13: Jansen J., On Competition and the Strategic Management of Intellectual Property in Oligopoly
- 2009/12: Hellwig M., Utilitarian Mechanism Design for an Excludable Public Good

published in: Economic Theory, vol. 2009, no. July 14, Berlin/Heidelberg, Springer, 2009.

- 2009/11: Weinschenk P., Persistence of Monopoly and Research Specialization
- 2009/10: Horstmann N., Ahlgrimm A., Glöckner A., How Distinct are Intuition and Deliberation? An Eye-Tracking Analysis of Instruction-Induced Decision Modes
- 2009/09: Lübbers T., Is Cartelisation Profitable? A Case Study of the Rhenish Westphalian Coal Syndicate, 1893-1913

2009/08: Glöckner A., Irlenbusch B., Kube S., Nicklisch A., Normann H., Leading with(out) Sacrifice? A Public-Goods Experiment with a Super-Additive Player

forthcoming in: Economic Inquiry, In Press.

- 2009/07: von Weizsäcker C., Asymmetrie der Märkte und Wettbewerbsfreiheit
- 2009/06: Jansen J., Strategic Information Disclosure and Competition for an Imperfectly Protected Innovation

forthcoming in: Journal of Industrial Economics, In Press.

2009/05: Petersen N., Abkehr von der internationalen Gemeinschaft? – Die aktuelle Rechtsprechung des US Supreme Court zur innerstaatlichen Wirkung von völkerrechtlichen Verträgen –

forthcoming in: Völkerrecht im innerstaatlichen Bereich, Vienna, facultas.wuv, In Press.

- 2009/04: Rincke J., Traxler C., Deterrence Through Word of Mouth
- 2009/03: Traxler C., Winter J., Survey Evidence on Conditional Norm Enforcement
- 2009/02: Herbig B., Glöckner A., Experts and Decision Making: First Steps Towards a Unifying Theory of Decision Making in Novices, Intermediates and Experts
- 2009/01: Beckenkamp M., Environmental dilemmas revisited: structural consequences from the angle of institutional ergonomics, issue 2009/01

#### Preprints 2008

- 2008/49: Glöckner A., Dickert S., Base-rate Respect by Intuition: Approximating Rational Choices in Base-rate Tasks with Multiple Cues
- 2008/48: Glöckner A., Moritz S., A Fine-grained Analysis of the Jumping to Conclusions Bias in Schizophrenia: Data-Gathering, Response Confidence, and Information Integration
- 2008/47: Hellwig M., A Generalization of the Atkinson-Stiglitz (1976) Theorem on the Undesirability of Nonuniform Excise Taxation
- 2008/46: Burhop C., The Underpricing of Initial Public Offerings in Imperial Germany, 1870-1896

forthcoming in: German Economic Review, In Press.

2008/45: Hellwig M., A Note on Deaton's Theorem on the Undesirability of Nonuniform Excise Taxation

forthcoming in: Economics Letters, In Press.

2008/44: Hellwig M., Zur Problematik staatlicher Beschränkungen der Beteiligung und der Einflussnahme von Investoren bei großen Unternehmen

published in: Zeitschrift für das gesamte Handelsrecht und Wirtschaftsrecht, vol. 172, pp. 768-787, 2008.

2008/43: Hellwig M., Systemic Risk in the Financial Sector: An Analysis of the Subprime-Mortgage Financial Crisis

published in: De Economist, no. 16.07.2009: Springer US, 2009.

published in: Jelle Zijlstra Lecture, no. 2008/5, Wassenaar, NL, Netherlands Institute for Advanced Study in the Humanities and Social Sciences, Institute of the Royal Netherlands Academy of Arts and Sciences, pp. 100, 2008.

- 2008/42: Glöckner A., Herbold A., Information Processing in Decisions under Risk: Evidence for Compensatory Strategies based on Automatic Processes
- 2008/41: Lüdemann J., Magen S., Effizienz statt Gerechtigkeit

forthcoming in: Zeitschrift für Rechtsphilosophie, In Press.

2008/40: Engel C., Die Bedeutung der Verhaltensökonomie für das Kartellrecht